Author’s response to reviews

Title: Phoyunnanin E inhibits migration of non-small cell lung cancer cells via suppression of epithelial-to-mesenchymal transition and integrin αv and integrin β3

Authors:

NAREERAT PETPIROON (petpiroon_bright@hotmail.com)

BOONCHOO SRITULARAK (pithi.c@chula.ac.th)

Pithi Chanvorachote (pithi_chan@yahoo.com)

Version: 2 Date: 29 Nov 2017

Author's response to reviews:

Response to reviewer

Please make the following revisions to your manuscript:

1. Please include the email addresses of all authors on the title page of your manuscript.

2. Please add a Conclusion section (heading) after the Discussion section of your main manuscript, in order to adhere to journal submission requirements.

3. Please include a "Declarations" section at the end of your manuscript. Contained within this section should be a list of abbreviations, as well as sections on 'Ethics approval and consent to participate', 'consent for publication', 'Availability of data and materials', 'Competing interests', 'Funding', 'Authors' Contributions' and 'Acknowledgements'. In the event that any of these sections are not relevant for your study, we would still ask you to include them, but to include a statement detailing that this information is not relevant. Further details on the information that should be included in your "Declarations" section, as well as how to format it, can be found in our submission guidelines (http://bmccomplementalternmed.biomedcentral.com/submission-guidelines/research-article).
4. Please delete your supplementary files. They are longer necessary at this stage of production.

Response: we have revised the manuscript as recommended.